Author's response to reviews

Title: Infection control and the burden of tuberculosis infection and disease in health care workers in China: a cross-sectional study

Authors:

Guang Xue He (heguangxue@chinatb.org)
Susan van den Hof (vandenhofs@kncvtbc.nl)
Marieke J van der Werf (vanderwerfm@kncvtbc.nl)
Guo Jie Wang (wgiwangguoie@163.com)
Shi Wen Ma (masw@hncdc.com.cn)
Dong Yang Zhao (zhaody@hncdc.com.cn)
Yuan Lian Hu (huyuanlian@foxmail.com)
Shi Cheng Yu (shicheng_yu@hotmail.com)
Martien W Borgdorff (m.w.borgdorff@amc.uva.nl)

Version: 9 Date: 27 October 2010

Author's response to reviews:

Dear The BMC Series Editorial Production Team,

Thank you very much for you found a mistake of reference 17 in our manuscript. I am sorry that I made this mistake when I revised the manuscript in September. In fact, the references in line 5, paragraph 2, page 7 should be [17, 18]. I have revised it.

Best wishes,

Guang Xue He